



BAISHIDENG PUBLISHING GROUP INC

8226 Regency Drive, Pleasanton, CA 94588, USA

Telephone: +1-925-223-8242 Fax: +1-925-223-8243

E-mail: bpgoffice@wjgnet.com <http://www.wjgnet.com>

Name of Journal: *World Journal of Clinical Infectious Diseases*

ESPS Manuscript NO: 20004

Manuscript Type: MINIREVIEWS

Answering Reviewers

There are no comments from reviewers on the content of the manuscript.